Dear Professors Chudek, Kontogeorgos, Ku, Peng, and Serati,

Thank you very much for your E-mail of June 18, 2023, with regard to our manuscript (#85872, entitled "Knee locking caused by osteochondroma of the proximal tibia adjacent to the pes anserinus: A case report") together with the comments from the reviewers. We herein resubmit our revised manuscript.

Our incorporation of the reviewers’ suggestions is as follows:

Response to Reviewer #1

Thank you for your comments, which have helped us improve our manuscript's contents. New text in the manuscript is presented in red font.

Comment 1: I am not satisfied with the differential diagnosis. Authors must and should mention the possibility of apophysitis (Osgood Schlatter syndrome) and other similar occurrences.

Response: We inserted the sentence about the differential diagnosis including apophysitis
Comment 2: In the manuscript there are mentioned two tables (Table 1 and Table 2) on page 11. However, although I searched several times, I could not find any table at all inside the word document of the manuscript. No tables were available as well in the documents in the online system. Kindly check and re-submit tables, possibly through embedding these inside the main document file.

Response: We apologize for our mistake not to embed two tables in the manuscript. We merged two tables, designated it as table 1, and embedded it in the main document (Page 22, line 10-Page 24).

Response to Reviewer #2

Thank you for your comments, which have helped us improve our manuscript's contents. New text in the manuscript is presented in red font.

Comment 1 and 2: 1. Tables 1 and 2 are not available in the manuscript. 2. References 16 and 17 are not mentioned in the text.

Response: We apologize for our mistake not to embed two tables in the manuscript. We merged two tables including references #20, 21 (former #16, 17), designated it as table 1, and embedded it in the main document (Page 22, line 10-Page 24)

Comment 3: I suggest authors add a list of abbreviations at the end of the manuscript.

Response: We added a list of abbreviations on Page 13, line 7-9.
We have had a native English-speaking editor review the manuscript again. We believe the manuscript has been improved satisfactorily and hope it will be accepted for publication in your journal, “World Journal of Clinical Cases”.

Yours sincerely,

Michiyuki Hakozaki, M.D., Ph.D.
Department of Orthopaedic Surgery,
Fukushima Medical University School of Medicine
1 Hikarigaoka, Fukushima-shi,
Fukushima 960-1295,
Japan
TEL: +81-24-547-1276
FAX: +81-24-548-5505
E-mail: paco@fmu.ac.jp